Author's response to reviews

Title: Nasal carriage of a single clone of community-acquired methicillin-resistant Staphylococcus aureus among kindergarten attendees in northern Taiwan

Authors:

Wen-Tsung Lo (drluoped@yahoo.com.tw)
Wei-Jen Lin (ndmcmdped@yahoo.com.tw)
Min-Hua Tseng (doc31089@yahoo.com.tw)
Jang-Jih Lu (jjl@ndmctsgh.edu.tw)
Mong-Ling Chu (mlchu@yahoo.com.tw)
Chih-Chien Wang (ndmcccw@yahoo.com.tw)

Version: 4 Date: 28 May 2007

Author's response to reviews: see over
Dear the BioMed Central Editorial Production Team

Thank you for your valuable recommendations for our manuscript. We have reviewed our article carefully and the final corrections are as follows:

1) Title page: We deleted “Full title”, “Abbreviated title”, “Keywords”, “Grant support…”, and “Potential financial conflicts…” as suggested (Page 1, Line 1-5).
2) We removed unnecessary capitalization from headings (Page 1, Line 1, 2).
3) We changed “Conclusion” to “Conclusions” (Page 4, Line 2; Page 14, Line 16).
4) We changed “Figure” to “Figure 1” (Page 11, Line 11; Page 27, Line 2).
5) Websites: We removed “(http://www.mlst.net)” from the text (Page 8, Line 10).
6) We would like to change “Table” to “Table 1” (Page 11, Line 2; Page 28, Line 1).
7) We would like to change “List of abbreviations used” to “Abbreviations used” (Page 16, Line 1).

We would like to take this opportunity to thank you for your most valuable, expert advice. We also greatly appreciate your patience in editing our article.

Sincerely yours,

Corresponding author: Dr. Chih-Chien Wang

Department of Pediatrics
Tri-Service General Hospital, National Defense Medical Center
No. 325, Cheng-Kung Road, Section 2, Nei-hu 114, Taipei, Taiwan.
Telephone: 886-2-87927025
Fax: 886-2-87927293
E-mail: ndmcccw@yahoo.com.tw